Author's response to reviews

Title: Epidemiologic Natural History and Clinical Management of Human Papillomavirus (HPV) Disease: A Critical and Systematic Review of the Literature in the Development of An HPV Dynamic Transmission Model

Authors:

Ralph P Insinga (ralph_isinga@merck.com)
Erik J Dasbach (erik_dasbach@merck.com)
Elamin H Elbasha (elamin_elbasha@merck.com)

Version: 4 Date: 17 July 2009

Author's response to reviews:

Dear Dr. Zauner,

We have addressed the remaining queries from the Editorial Team as follows:

(1) Material previously in the Appendix has been moved to Table 1 within the body of the manuscript.

(2) A new table (Table 2) has been added, with a detailed summary of the methodological criteria used to select studies to inform the parameters described in the paper.

(3) Author contribution and declaration of competing interest statements have been added at the conclusion of the paper.

Best regards,

Ralph Insinga